

Archive ouverte UNIGE

https://archive-ouverte.unige.ch

Article scientifique

Article

2019

Published version

Open Access

This is the published version of the publication, made available in accordance with the publisher's policy.

Giving birth in Switzerland: a qualitative study exploring migrant women's experiences during pregnancy and childbirth in Geneva and Zurich using focus groups

Sami, Jana; Quack Lötscher, K C; Eperon, Isabelle; Gonik, Léa; Martinez De Tejada Weber, Begona; Epiney, Marie-Christine; Schmidt, Nicole Christine

How to cite

SAMI, Jana et al. Giving birth in Switzerland: a qualitative study exploring migrant women's experiences during pregnancy and childbirth in Geneva and Zurich using focus groups. In: Reproductive health, 2019, vol. 16, n° 1, p. 112. doi: 10.1186/s12978-019-0771-0

This publication URL: https://archive-ouverte.unige.ch/unige:143375

Publication DOI: <u>10.1186/s12978-019-0771-0</u>

© The author(s). This work is licensed under a Creative Commons Attribution (CC BY) https://creativecommons.org/licenses/by/4.0

RESEARCH Open Access

Giving birth in Switzerland: a qualitative study exploring migrant women's experiences during pregnancy and childbirth in Geneva and Zurich using focus groups



J. Sami¹, K. C. Quack Lötscher², I. Eperon³, L. Gonik³, B. Martinez de Tejada^{1,3}, M. Epiney^{1,3} and N. C. Schmidt^{3*}

Abstract

Background: Migrant mothers in high-income countries often encounter more complications during pregnancy, delivery, and the postpartum period. To enlighten health care providers concerning potential barriers, the objective of this study was to explore positive and negative experiences with maternal health services in the University Hospitals of Geneva and Zurich and to describe barriers to maternity services from a qualitative perspective.

Methods: In this qualitative study, six focus groups (FGs) were conducted involving 33 women aged 21 to 40 years. All FG discussions were audio-recorded and later transcribed. Data were analysed using a thematic analysis approach assisted by the Atlas.ti qualitative data management software.

Results: Positive experiences included not only the availability of maternity services, especially during emergency situations and the postpartum period, but also the availability of specific maternity services for undocumented migrants in Geneva.

Negative experiences were classified into either personal or structural barriers. On the personal level, the main barriers were a lack of social support and a lack of health literacy, whereas the main themes on the structural level were language barriers and a lack of information.

Conclusion: Structural adaptation is necessary to meet the needs of the extremely diverse population. The needs include (1) the provision of specific information for migrant women in multiple languages, (2) the availability of trained interpreters who are easily accessible to health care providers, (3) specifically trained nurses or social assistance providers to guide migrants through the health system, and (4) a cultural competence-training programme for health care providers.

Keywords: Maternal health, Pregnancy, Migrant woman, Qualitative study, Focus group

³Obstetrics Unit Department of Obstetrics and Gynecology, Geneva University Hospitals, Geneva, Switzerland Full list of author information is available at the end of the article



^{*} Correspondence: nicole.ch.schmidt@gmail.com

Sami et al. Reproductive Health (2019) 16:112 Page 2 of 9

Plain English summary

Migrant mothers in high-income countries often encounter more complications during pregnancy, delivery, and the postpartum period. The objective of this study was to explore positive and negative experiences with maternal health services in the University Hospitals of Geneva and Zurich and to describe barriers to maternity services. The study used a qualitative approach, involving 33 women aged 21 to 40 years in six focus groups. The data was analyzed, and common themes were identified.

Positive experiences included the availability of maternity services, especially during emergency situations, but also the availability of specific maternity services for undocumented migrants in Geneva.

Negative experiences were a lack of social support, a lack of health literacy, but also language barriers and a lack of information.

In conclusion, the study suggested the following four structural adaptations:

- (1) the provision of specific information for migrant women in multiple languages
- (2) the availability of trained interpreters who are easily accessible to health care providers
- (3) specifically trained nurses or social assistance providers to guide migrants through the health system
- (4) a cultural competence-training programme for health care providers

Introduction

In 2017, 258 million individuals worldwide were living in a country other than their birth country [1]. The International Organization of Migration (IOM) defines a migrant as "a person who who moves away from his or her place of usual residence, whether within a country or across an international border, temporarily or permanently, and for a variety of reasons", regardless of the person's legal status and whether the movement is voluntary or involuntary [2].

Almost one-third of the world's international migrants (78 million) lived in Europe in 2017 [1]. The same year, the canton of Geneva and the city of Zurich reported that 40.2 and 32.4% of their respective residents had a nationality other than Swiss, and these values were higher than the Swiss average of 25% [3–6].

Adapting to a new country can raise important challenges, and the previous literature has described health disparities between migrants and host populations [7, 8]. Among other issues, migrants have been reported to be more susceptible to suffering from diabetes, mental illness and maternal and child health problems [9].

Furthermore, studies have found that migrant women encounter more complications during pregnancy, childbirth and the postpartum period [10, 11]. In Switzerland, these observations have been confirmed by quantitative studies that have shown a higher maternal mortality rate among non-Swiss than among Swiss women and higher rates of unwanted pregnancies and delayed prenatal care in undocumented women [12–14].

Factors leading to these health disparities are difficult to disentangle, and multiple reasons have been proposed, including language barriers, a lack of health literacy and a lack of information [15]. A Swiss qualitative study by Bollini and colleagues explored maternity experiences among Turkish and Portuguese women in Switzerland and reported that these groups faced stressful situations that negatively affected their pregnancies [16]. Furthermore, a poorer level of satisfaction with maternity services may be a barrier for utilization of these services by pregnant migrants [15, 16].

To reduce maternal health disparities, their causes must be understood. Therefore, a qualitative study design was chosen with the aim of describing the experiences of migrant women with pregnancy and maternity services at two main hospitals of Geneva and Zurich and to identify specific barriers for health care accessibility.

Methodology

Study design and sampling

Between September 2015 and February 2016, a qualitative study using focus groups was conducted with migrant women who were either pregnant or in their first year after childbirth in Geneva and Zurich. Focus group are a common method in qualitative health research to capture insights related to the ways people perceive and interpret their surroundings [17, 18]. Each focus group (FG) was led by two researchers. While one researcher facilitated the discussion based on a semi-structured questionnaire, the other observed group dynamics and body language. This approach has been chosen as an adequate method because it reduces for example the influence of the researcher [19].

Discussion topics included 1) health during pregnancy, 2) experience with the Swiss health care system during pregnancy and childbirth, 3) perceived barriers, and 4) suggestions for improvement of the current situation.

Multiple recruitment strategies to access women during pregnancy or in their first year postpartum were used, including personal contact during antenatal care visits, contacting migrant organizations by email or telephone and the snowball method.

Six focus groups (FGs) were conducted with a total of 33 women, who had moved to Switzerland. Migrant women included in the study moved away from their home countries to one or more of the following reasons: Sami et al. Reproductive Health (2019) 16:112 Page 3 of 9

- Woman had left her country involuntarily either due to war, conflict or economic pressure.
- Woman lived in Switzerland without a legal status.
- Woman had recently arrived.

For women having migrated for one or more of the mentioned reasons, difficulties in accessing health services in Switzerland have been described in the literature [12, 14].

Additional characteristics of the FGs and participants are noted in Tables 1 and 2.

Data collection and analysis

The study was conducted in the Services of Obstetrics of the University Hospitals of Geneva and the Department of Obstetrics of the University Hospital Zurich. Geneva and Zurich are two of the most important cities for immigration in Switzerland, and therefore the experiences of migrant women during pregnancy are considered of high importance for health care providers [5, 6]. The Ethical Commissions in Geneva and Zurich approved the study.

The FGs were led by one of the main investigators (NCS or KQL). After obtaining written consent from each participant, the FGs were conducted in private rooms in the University hospitals in either English or the maternal language of the participants with the assistance of a skilled interpreter who was accepted by the group.

The FG discussions were recorded, transcribed and translated into English. The transcripts were analysed using ATLAS.ti. CAQDAS and coded by two different members of the research team independently to avoid subjectivity. After individual coding, both researchers identified the main topics and sub-topics common to every group and outlined the women's experiences with the Swiss health care system as well as the identified barriers.

Our study adopted the conceptual framework of Higginbottom and colleagues [20]. In this study, published

in 2016, experiences of immigrant women with maternity care services in rural and urban Alberta, Canada, were investigated. Barriers were structured at the patient (personal) and maternity services (organizational) levels. The conceptual framework of Higginbottom et al. was chosen because identifying barriers especially on the organizational level enables health care institutions to make adjustments in order to provide equitable health-care services for an increasingly diverse population [20].

Results

The 33 participants were all born outside of Switzerland and had been in Geneva or Zurich for an average of 5.5 years. They originated from nine different countries, were on average 29 years old (range 21 to 40 years) and most were married (89%). Their education level was good, with more than 80% having completed high school. Nine out of 15 participants in Geneva and one participant in Zurich were living in Switzerland without a legal status and didn't have a health insurance (Table 2).

The results are organized into two parts. In the first part, migrant women's good experiences during maternity care are outlined. In the second part, the six major themes influencing the participant's access to maternity care services are described and classified into either barrier on the patients or maternity service level.

Good maternity care experiences

Pregnancy check-ups and availability of services in case of perceived emergency situations

The participants appreciated the availability of health care services and the possibility of calling health care providers (HCPs) with questions or perceived emergency situations. H. from Eritrea stated: "I would say it's better in Switzerland, because no matter what time it is, when you feel sick, you can call a doctor and you get an appointment for the next day right away."

Table 1 Focus group characteristics

FG	Number of participants (total = 33)	Number of women (pregnant or postpartum)	Country/Countries of origin	Language of the interview (assisted by an interpreter)	Location
1	2	2 postpartum	Portugal	Portuguese (interpreter)	Zurich
2	4	2 pregnant	Peru, USA, Germany	English	Geneva
		2 postpartum			
3	5	2 pregnant	Dominican Republic, Peru, Bolivia	Spanish	Geneva
		3 postpartum			
4	6	6 pregnant	Brazil	Portuguese (interpreter)	Geneva
5	4	2 pregnant	Eritrea	Tigrinya (interpreter)	Zurich
		2 postpartum			
6	12	7 pregnant	Bangladesh	Bengali (interpreter)	Zurich
		5 postpartum			

Sami et al. Reproductive Health (2019) 16:112 Page 4 of 9

Table 2 Participant characteristics

	n = 33 (%)
Women 21–40 years of age	33 (100%)
Mean age in years (SD)	29.2 (5.5)
Mean years in Switzerland	
-less than 4 years	18 (54.5%)
-more than 4 years	15 (45.5%)
Religion	
Christian	15 (45.5%)
Muslim	11 (33.3%)
Other	2 (6.0%)
No answer	5 (15.2%)
Civil status	
Single	2 (6.1%)
Married or in partnership	29 (87.9%)
Separated or divorced	1 (3.0%)
No answer	1 (3.0%)
Education	
Less than 12 years	3 (9.1%)
High school (=12 years)	18 (54.5%)
Education > 12 school years	10 (30.3%)
No answer	2 (6.1%)
Children	
Expecting first child	7 (21.2%)
Already had children	26 (78.8%)
Work situation	
Full time or part time	10 (30.3%)
Not working	20 (60.6%)
Other (student or inability to work)	2 (6.1%)
No answer	1 (3.0%)
Health insurance	
Yes	17 (51.5%)
No	10 (30.3%)
No answer	6 (18.2%)

Furthermore, most participants felt reassured by the number of pregnancy check-ups in comparison to those in their home country. S., who was 6 months pregnant and had given birth to two children in Brazil, said: "I think here is better than there, because here they really make efforts to know how we feel, how we are, etc. In Brazil, I had an ultrasound during the first month of my pregnancy, and then at the end of it, when I was about to give birth. On the other hand, here I am six months pregnant and this is already the fifth ultrasound, because they want to be sure about the baby, how is the baby, if everything is alright ...".

Postpartum period

In Switzerland in 2015 and 2016, all women with health care insurance, which is mandatory for legal residents, were entitled to a daily visit with a midwife until the tenth day after birth, even when at home. The midwife helps to care for the women, provides support with breastfeeding and answers their questions.

Most participants who had previously given birth in Switzerland appreciated the postpartum care services provided by the midwives. S. from Brazil stated: "Once you give birth, a midwife comes. In Brazil, it is different. When I had my baby, I was 16 years old and I had to do everything by myself."

Specific services for women without legal status

Nine out of 15 women in the FGs in Geneva were without legal status. The women without legal status appreciated a service available only at the Geneva site that helped undocumented migrants' access medical care. It also facilitated access for undocumented pregnant women to the University Hospital, where a specific antenatal care consultation was provided.

BARRIERS on the patient side or the maternity service level Patient side

Lack of social support from family members In nearly all of the FGs, the women described a lack of social support and a feeling of loneliness. S.B. from Eritrea explained how she felt: "When you have your parents with you, they are there to help because they feel responsible for you. Therefore, I find it harder in Switzerland compared to Eritrea."

Lack of health literacy Health literacy has been defined by the World Health Organization as « the cognitive and social skills which determine the motivation and ability of individuals to gain access to, understand and use information in ways which promote and maintain good health » [21]. Women coming from countries with less developed health systems and lower education sometimes misinterpreted medical procedures due to a lack of information, but also the ability to understand the information. They were worried about the consequences of medical examinations. S.B. from Eritrea said: "Yes! I mean maybe it's not intentional. It could be that when the doctor puts instruments inside of you to check the baby, he or she accidentally damages the foetus, which in turn may lead to a miscarriage ...".

Furthermore, during several FGs, divergent perceptions and misunderstandings with respect to the first trimester screening were described. The first trimester screening test, which is routinely offered in Switzerland

Sami et al. Reproductive Health (2019) 16:112 Page 5 of 9

to all women between 11 to 14 weeks of pregnancy to assess the risk for trisomy 13, 18 and 21 in the unborn, includes a blood test combined with an ultrasound. However, misunderstanding of these procedures led to anxiety and in some cases to refusal. V., a Brazilian mother, explained the reason for her refusal: "Yes, but she [the doctor] didn't explain to me right how this exam would be, and I thought it was the one with the needle, that's why I said no and asked her if I could do it later."

The calculation of risk was also difficult to understand for several participants. As C. from Germany explained: "I said yes, for me if the child is definitely sick I want to know and if he's definitely not sick I want to know, but everything in between, I don't want to know, I'd rather just have it and deal with it afterwards."

Other women reported not wanting to undergo this test because the result would not change anything, but they felt pressured to stop their pregnancy if the risk for a medical pathology was elevated. C.E. from Brazil said: "Here, when you say I am pregnant, they say: here [in Switzerland], you can abort. It's some sort of pressure for us. That's what I find different. For example, in Brazil, for our culture and our people, if God sends us a baby that has some problems, we will still have the baby. Here, seems that there is a pressure if the baby has problems, you can just remove it."

Perceived discrimination During the FGs, several women described a feeling of discrimination, especially if they were living in vulnerable situations in Switzerland, such as difficult social or financial conditions, or if they were living without a legal status. L.S. from Brazil felt pressured by the social worker at the hospital to stop her pregnancy: "No, I think they do that because we are not married. In my case, I felt it happened because I was alone, because they suggested it several times... I think it's because of a financial matter."

Additionally, a feeling of discrimination was reported during pregnancy check-ups. In University Hospitals, where most of the migrant women went for their maternity care, medical students can attend to consultations with the patient's permission. Several women from different FGs explained that they thought doctors allowed students to attend the consultation mainly because they were migrants.

Maternity services level

Language barrier and lack of information The main barriers at the maternity service level were language barriers and lack of information. Not being able to express oneself properly because of language differences negatively influenced the doctor-patient relationship and led to frustration or worries. Participants expressed the need to have more information about pregnancy in Switzerland available either in English or in their native language, because they were not able to ask questions or did not understand the answers. K. from Peru explained: "I don't know how it is here, what to eat or not eat, what should I drink… if I'll understand the doctor, if they'll understand me, because I don't speak French".

Administrative procedures Migrant women described difficulties understanding the administrative procedures of the Swiss health care system. These difficulties were in some situations closely related to a lack of health literacy on the patient side or a lack of information. N. from Peru noted: "I think the doctor assumes that you know everything. I didn't know that maternity services were covered 100%. I found out when my first bill came. Probably, there is information, but I don't know where to find it."

In addition, women described that scheduling medical appointments was sometimes difficult and time consuming, especially for pregnant women who did not speak the native language. One woman who resided in Geneva and did not speak French explained: "But you got to find it (the information) and know it's there and kind of access it. It takes a lot of effort to cross that threshold. It takes a lot of energy and that's the other thing: you're tired all the time! You have to call someone that you don't know, call them in French and try to schedule an appointment."

Lack of continuity of care For several participants, not being able to stay with the same HCP during the whole pregnancy was difficult. This situation is especially the case for women with basic health insurance, because they cannot give birth with their doctor if he does not work in a public hospital. N. from Peru explained: "The only problem is that he (her gynaecologist) works only at a private clinic, not at the public hospital, so I already said goodbye to him about 2 weeks ago."

Furthermore, women expressed their concerns about the constant change of HCPs once they arrived at the public hospital, as R. from Bolivia explained: "I think it's never the same doctor from one ultrasound to the other. That is difficult."

Discussion

The current study is, to our knowledge, among the first in Switzerland with the aim of understanding experiences with maternity care services among pregnant migrants in two major cities for Switzerland from a qualitative perspective. Two main themes emerged in respect to the positive experiences: the availability of services in perceived emergency services and the postpartum care services. Six main themes were identified when women described their negative experiences. Those were categorized either in barriers on the personal or structural (maternity level) side: lack of social

Sami et al. Reproductive Health (2019) 16:112 Page 6 of 9

support from family member, lack of health literacy, perceived or real discrimination, language barriers and lack of information, administrative procedures and lack of continuity of care.

In general, our results in respect to perceived barriers are fairly consistent with previous national and international literature. However, until now few studies have described positive experiences. The results of our study showed that migrant women highly appreciated the availability of maternity services, especially in perceived emergency situations, and the postpartum care offered by midwives at home. In particular, women who had previously given birth in countries where routine postpartum care did not exist highlighted this service.

Furthermore, undocumented women in Geneva were aware of and appreciated the dedicated services provided for them by specific trained nurses and social assistants at the maternity hospital. This service is especially important, because the previous literature has reported that undocumented pregnant women have poorer access to pregnancy screening and check-ups [12, 22]. A recent paper by Balaam et al. showed that caring relationships (including those in health care) might help migrant women access maternity care [23]. Therefore, the availability of services such as in Geneva that are adapted to women's needs may allow equity in accessing maternity care for migrants, including undocumented patients.

However, besides the positive experiences, several negative experiences and barriers were described. One important barrier mentioned by nearly all participants was the lack of social support. Most migrants explained that in their home countries women were supported during pregnancy, childbirth and in the first weeks after childbirth by their parents or family members. As parents or family members remained mostly in the home country, women described anxiety and loneliness. These findings are in contrast to a previous study conducted in Geneva among migrant women who did not mention a lack of social support [24]. One possible explanation might be the length of residency in the recipient country. Over 50% of the women in the current study had been in Switzerland for less than 4 years, whereas in the previous study over 60% of the participants had lived in Geneva for more than 4 years. Length of stay in a new country has been mentioned has an important factor for acculturation that allows among others the understanding of a new health systems [25]. Furthermore, this study included women during pregnancy or in the first year after childbirth. Family members and close friends might be perceived as especially important during this period.

Lastly, the type of migration experience needs to be considered. Several women had left their home country due to war or conflict. Women who have experienced war or even sexual violence are more at risk to suffer from physical or mental health problems [26]. In cases of involuntary migration, women might miss the social support from their family members even more. The same holds true for undocumented migrants, who participated in the study and for whom family unification was not possible. Health care professionals should be aware of this barrier, because a lack of social support has been linked to a higher risk of postpartum depression [20, 23, 27–29]. Even though the assessment of postpartum depression was not part of our study, evaluation by health professionals and the provision of support becomes especially important [23, 27, 28].

A second important barrier on the patient side was the lack of health literacy. "Health literacy" has been defined by the World Health Organization as "the cognitive and social skills which determine the motivation and ability of individuals to gain access to, understand and use information in ways which promote and maintain good health" [21]. Women often lacked information about the Swiss health care system, including administrative procedures, such as scheduling medical appointments. These observations align with findings from previous studies in maternity care [15, 20, 24]. Even if maternity services were equally available for all women, in practice the services were often not accessible to migrant women due to a lack of awareness about their existence [20]. Additionally, the institutional culture of maternity services has been reported to have been mainly designed for those who understand and can negotiate the system. Therefore, people who recently have arrived and are not familiar with the system face a range of barriers [30, 31].

Interestingly, none of the undocumented pregnant women in Geneva mentioned difficulties in scheduling appointments or the lack of continuity of care as a barrier. A possible explanation may be the dedicated services for them mentioned above, which guide undocumented migrant women through the system and provide antenatal and postpartum care. These observations were in accordance with those from a recent study in Switzerland, which assumed that those services might be helpful for migrants who had arrived recently and were not yet familiar with the system [24].

In our study, a lack of health literacy represented a source of worries for pregnant women, because they were sometimes not used to the medical procedures routinely offered in Switzerland. The consequent misunderstandings negatively influenced the patient-doctor relationship and were previously described as an important barrier to the use of health services [9, 15]. This finding is particularly important, since some HCPs are not aware of the lack of health literacy as a barrier and fail to explain medical procedures adequately to their patients. Therefore, increasing HCP awareness of women's perceptions of health

Sami et al. Reproductive Health (2019) 16:112 Page 7 of 9

services through a cultural competence training programme in necessary, as has already been mentioned previously [30, 32]. Aside from the importance of setting up culturally sensitive care, the misunderstandings towards first trimester screening also show the importance of providing clear information to migrant women, as discussed further.

Furthermore, the lack of health literacy in our study was closely related to the lack of information, which might limit access to health care [33]. The inadequate provision of informative material was one of the main barriers at the organizational level. Guaranteeing "the same access to information to the native and migrant populations" has been reported by Bischoff as an essential intervention to promote health for migrant patients [30]. This finding is in accordance with our study and previous studies that have highlighted the need for migrant women to access information in their first language [31, 34]. Indeed, because recommendations during pregnancy may vary from one country to another, every woman should receive information resources for the host country provided by the HCP in her first or any familiar language [14].

The final barrier, which influences directly women's health literacy, was suboptimal communication between HCPs and migrant women. This barrier is mainly a result of insufficient language proficiency on the patient side, which is necessary to understand the information provided by and to communicate successfully with the HCP. As highlighted by previous studies, language is one of the main barriers for health care access and satisfaction with on-going care as misunderstanding leads to lack of information and stressful situations [9, 15, 20]. Our study confirmed previous results showing that women relied mostly on ad hoc interpreters, such as family members, to overcome this barrier. This issue is especially important, because the use of ad hoc interpreters has been proven to lead to poor quality communication and a lack of confidentiality [33, 35, 36]. None of the participants in our study seemed to be aware of the provision for professional interpreters free of charge at the University Hospital Geneva and University Hospital Zurich. Therefore, efforts should be undertaken by health institutions to increase awareness of the availability of such services [37].

Furthermore, health care institutions need to address the barriers described on the structural (maternity) level and raise the awareness of health service providers in respect to those on the patient side. Therefore, the following strategies are suggested to reduce some of the most important barriers at the maternity level: (1) specific information should be provided for migrant women in multiple languages and actively delivered, (2) the availability of trained interpreters must be highlighted and made easily accessible to HCPs to

overcome language barriers, (3) the use of professional social accompaniment, such as help with administrative procedures, should be considered, and (4) a cultural competence training programme for HCPs should be implemented with the aim of improving communication, increasing awareness of migrant women's perceptions of health care and preventing discrimination.

Limitations of the study

Although this qualitative study was among the first to explore barriers to maternity care services in the canton of Geneva and the city of Zurich, it faced several limitations.

First, the qualitative approach using focus groups covers the range of topics considered important by the participants but cannot be generalised. Also, the methodology of a focus group design might have prevented some participants to express their honest and personal opinion. However, to limit this influence, small groups with participants from the same social background were chosen. Furthermore, FGs were conducted in a private and confidential setting.

Second, migrant women present a very heterogenous group with different needs and experiences. Even if the study assessed for example the length of the stay in Switzerland and the legal status, future studies should include also information about the type of migration experience (forced vs. voluntary) or the length of residence in the recipient country [38]. Also, as already outlined, migrant women who have been forced to leave their home country, might have different expectations, experiences and health problems compared to those having migrated for economic opportunities. Additionally, the authors of this article do not belong to an ethnic minority group. Even though the authors tried to reduce those barriers by discussing them with representatives from migrant communities, an interviewer bias from a western perspective cannot be excluded.

Finally, understanding the experiences and perceived barriers of Swiss pregnant women or those in their first year postpartum would have been helpful to clarify whether organizational barriers are specific to migrant women or are also present for Swiss women.

Conclusion

The Swiss health care system is meant to provide health care for everyone irrespective of their origin and legal status. Even if this goal holds true for maternity health services in theory, the barriers described in our study demonstrate that maternity health services are not always accessible. Although qualitative results cannot be generalized, we believe that our results are confirmed by the national and international literature. Therefore, reducing these barriers may be beneficial at personal and institutional levels and may support health system strategies to improve health equity.

Sami et al. Reproductive Health (2019) 16:112 Page 8 of 9

Barriers are strongly influenced by personal factors, but organizational barriers also exist. In our opinion, the latter are especially important, because policy makers and HCPs can act to overcome them. Strategies to reduce some of the most important barriers at the maternity level should focus on language barriers and the improvement of health literacy, but also address perceived discrimination and administrative procedures.

Acknowledgements

The authors acknowledge the participating women and the assisting interpreters who dedicated their time and shared their experiences. Additionally, we especially thank the midwives Lea Gonik, Patricia Sala and Marina Corteasao, who supported the recruitment of the participants in Geneva.

Authors' contribution

NCS, as the head investigator, developed the main idea, helped in all phases of the research, and participated in writing the draft and finalizing the manuscript. JS assisted in several focus groups, analysed and interpreted most of the qualitative data and wrote the draft of the manuscript. IE, LG, JS, ME, MB and KQL assisted with the recruitment of the study participants and provided essential comments on the final manuscript. All authors read and approved the final manuscript.

Funding

The study was part of a career funding granted by the Swiss National Foundation and provided part of the salary for the main investigator NCS. The salary of the main researcher NCS was complemented by a research grant from Geneva University. Furthermore, additional research costs were covered by the Federal Office of Health, Berne (Switzerland).

Availability of data and materials

The datasets (transcripts) generated and analyzed during the current study are not publicly available due to the sensitivity of the data, but summaries of the transcripts are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

Informed consent was obtained from all participants with clearly stated objectives of the research. The participants were also assured of the anonymity and confidentiality of the information and were informed of their right to participate or refuse participation. The Ethical Board of the canton of Geneva approved the study (CER 14–095). The datasets generated during and/or analysed during the current study are not publicly available due to the sensitivity of the data.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Faculty of Medicine, University of Geneva, Geneva, Switzerland. ²Clinic of Obstetrics, University Hospital Zurich, Zurich, Switzerland. ³Obstetrics Unit Department of Obstetrics and Gynecology, Geneva University Hospitals, Geneva, Switzerland.

Received: 29 January 2019 Accepted: 4 July 2019 Published online: 22 July 2019

References

 United Nations. Department of economic and social affairs, population division (2017): International Migration Report; 2017. Highlights (ST/ESA/SER. A/404). http://www.un.org/en/development/desa/population/migration/ publications/migrationreport/docs/MigrationReport2017_Highlights.pdf

- Glossary on Migration. 2019 International Organization for Migration (IOM) ISSN 1813-2278. https://publications.iom.int/system/files/pdf/iml_34_ glossary.pdf.
- Die Bevölkerung der Schweiz 2017. Bundesamt für Statistik. 2018. Available from: https://www.bfs.admin.ch/bfs/de/home/statistiken/bevoelkerung/ stand-entwicklung/bevoelkerung.assetdetail.6606496.html.
- Migration Report 2017. State Secretariat for Migration (SEM). https://www.sem.admin.ch/dam/data/sem/publiservice/berichte/migration/migrationsbericht-2017-e.pdf.
- Office cantonal de la statistique (OCSTAT). Bilan et état de la population du canton de Genève en 2017: Informations statistiques; 2018. Available from: https://www.ge.ch/statistique/tel/publications/2018/ informations_statistiques/autres_themes/is_population_06_2018.pdf. Accessed 23 Sept 2018.
- Anteil Ausländische Bevölkerung, 2017, https://www.stadt-zuerich.ch/prd/ de/index/statistik/themen/bevoelkerung/nationalitaet-einbuergerungsprache/anteil-auslaendische-bevoelkerung.html Accessed 23 Sept 2018.
- Marmot M, Allen J, Bell R, Bloomer E, Goldblatt P. Consortium for the European review of social determinants of health and the health divide. WHO European review of social determinants of health and the health divide. Lancet. 2012;380(9846):1011–29.
- WHO. The world health report 1998 life in the 21st century: a vision for all. World Health Organization 1998. Available from: http://www.who.int/whr/1 998/en/.
- Keygnaert I, Ivanova O, Guieu A, Van Parys AS, Leye E, Roelens K. What is the evidence on the reduction of inequalities in accessibility and quality of maternal health care delivery for migrants? a review of the existing evidence in the WHO European region. WHO health evidence network synthesis reports. WHO Regional Office for Europe: Copenhagen; 2016.
- Almeida LM, Caldas J, Ayres-de-Campos D, Salcedo-Barrientos D, Dias S. Maternal healthcare in migrants: a systematic review. Matern Child Health J. 2013;17(8):1346–54.
- Gagnon AJ, McDermott S, Rigol-Chachamovich J, Bandyopadhyay M, Stray-Pedersen B, Stewart D, et al. International migration and gestational diabetes mellitus: a systematic review of the literature and meta-analysis. Paediatr Perinat Epidemiol. 2011;25(6):575–92.
- Wolff H, Epiney M, Lourenco AP, Costanza MC, Delieutraz-Marchand J, Andreoli N, et al. Undocumented migrants lack access to pregnancy care and prevention. BMC Public Health. 2008;8:93.
- Bollini P, Wanner P, Pampallona S. Trends in maternal mortality in Switzerland among Swiss and foreign nationals, 1969-2006. Int J Public Health. 2011;56(5):515–21.
- Bollini P, Wanner P. Santé reproductive des collectivités migrantes: disparités de risques et possibilités d'intervention: SFM Institut; 2006. Available from: https://libra.unine.ch/Publications/Par-ann%E9e/Y-2 006/14103.
- Scheppers E, van Dongen E, Dekker J, Geertzen J, Dekker J. Potential barriers to the use of health services among ethnic minorities: a review. Fam Pract. 2006;23(3):325–48.
- Bollini P, Stotzer U, Wanner P. Pregnancy outcomes and migration in Switzerland: results from a focus group study. Int J Public Health. 2007; 52(2):78–86.
- Minkler M, Wallerstein N, editors. Communication Based Participatory Research for Health: From Process to Outcomes. 2nd edn. San Francisco: Wiley; 2008.
- Merry L, Clausen C, Gagnon AJ, Carnevale F, Jeannotte J, Saucier J-F, et al. Improving qualitative interviews with newly arrived migrant women. Qual Health Res. 2011;21(7):976–86.
- Frisina A. Focus groups in migration research: a forum for "public thinking"?
 In: Zapata-Barrero R., Yalaz E. (eds) Qualitative Research in European Migration Studies. IMISCOE Research Series. Springer, Cham. 2018.
- Higginbottom GM, Safipour J, Yohani S, O'Brien B, Mumtaz Z, Paton P, et al. An ethnographic investigation of the maternity healthcare experience of immigrants in rural and urban Alberta, Canada. BMC Pregnancy Childbirth. 2016:16:20.
- World Health Organization. Health promotion. Track 2: health literacy and health behaviour. 2009. http://www.who.int/healthpromotion/conferences/ 7gchp/track2/en/. Accessed 28 May 2019.
- Munro K, Jarvis C, Munoz M, D'Souza V, Graves L. Undocumented pregnant women: what does the literature tell us? J Immigr Minor Health. 2013;15(2):281–91.

Sami et al. Reproductive Health (2019) 16:112 Page 9 of 9

- Balaam MC, Akerjordet K, Lyberg A, Kaiser B, Schoening E, Fredriksen AM, Ensel A, Gouni O, Severinsson E, et al. A qualitative review of migrant women's perceptions of their needs and experiences related to pregnancy and childbirth. J Adv Nurs. 2013;69(9):1919–30.
- Schmidt NC, Fargnoli V, Epiney M, Irion O. Barriers to reproductive health care for migrant women in Geneva: a qualitative study. Reprod Health. 2018;15(1):43.
- David M, Borde T, Brenne S, Ramsauer B, Hinkson L, Henrich W, Razum O, Breckenkamp J. Obstetric care quality indicators and outcomes based on the degree of acculturation of immigrants-results from a cross-sectional study in Berlin. Arch Gynecol Obstet. 2018;297(2):313–22. https://doi.org/10.1 007/s00404-017-4574-7 Epub 2017 Oct 25.
- Sönmez E, Jesuthasan J, Abels I, Nassar R, Kurmeyer C, Schouler-Ocak M.
 Study on female refugees A representative research study on refugee
 Sönmez E, Jesuthasan J, Abels I, Nassar R, Kurmeyer C, Schouler-Ocak M.
 Study on female refugees A representative research study on refugee. Eur Psychiatry. 2017;41:S251. https://doi.org/10.1016/j.eurpsy.2017.02.038.
- Schmied V, Black E, Naidoo N, Dahlen HG, Kiamputton P. Migrant women's experiences, meanings and ways of dealing with postnatal depression: a meta-ethnographic study. PLoS One. 2017;12(3):e0172385.
- Collins CH, Zimmermann C, Howard LM. Refugee, asylum seeker, immigrant women and professionals. Cochrane Database Syst Rev. 2014;(5):CD009405.
- Falah-Hassani K, Shiri R, Vigod S, Dennis C-L. Prevalence of postpartum depression among immigrant women: a systematic review and metaanalysis. J Psychiatr Res. 2015;70:67–82.
- Bischoff A. Caring for migrant and minority patients in European hospitals. Université de Neuchâtel. 2006. Report No.: 43. Available from: http://doc.rero.ch/record/6438.
- Reitmanova S, Gustafson DL. They can't understand it': maternity health and care needs of immigrant Muslim women in St. John's, Newfoundland. Matern Child Health J. 2008;12(1):101–11.
- Horvat L, Horey D, Romios P, Kis-Rigo J. Cultural competence education for health professionals. Cochrane Database Syst Rev. 2014;(5):CD009405.
- Mangrio E, Sjörgen FK. Refugees' experiences of healthcare in the host country: a scoping review. BMC Health Serv Res. 2017;17:814.
- 34. Courtright C. Health information-seeking among Latino newcomers: an exploratory study. Inf Res. 2005;10(2). Available from: http://www.informationr.net/ir/10-2/paper224.html
- 35. Flores G, Laws MB, Mayo SJ, Zuckerman B, Abreu M, Medina L, et al. Errors in medical interpretation and their potential clinical consequences in pediatric encounters. Pediatrics. 2003;111(1):6–14.
- Elderkin-Thompson V, Silver RC, Waitzkin H. When nurses double as interpreters: a study of Spanish-speaking patients in a US primary care setting. Soc Sci Med. 2001;52(9):1343–58.
- Bischoff A, Hudelson P. Communicating with foreign language–speaking patients: is access to professional interpreters enough? J Travel Med. 2010;17(1):15–20.
- Balaam MC, Haith-Cooper M, Pařízková A, Weckend MJ, Feming V, Fossalu T, Vržina SŠ. A concept analysis of the term migrant women in the context of pregnancy. Int J Nurs Pract. 2017;23:e12600.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

